Author’s response to reviews

Title: A randomized, double-blind, positive-controlled, prospective, dose-response clinical study to evaluate the efficacy and tolerability of an aqueous extract of Terminalia bellerica in lowering uric acid and creatinine levels in chronic kidney disease subjects with hyperuricemia.

Authors:

Usharani Pingali (ushapingali@yahoo.com)
Chandrasekhar Nutalapati (csnpru@gmail.com)
Niranjan Koilagundla (dr.niranjandm@gmail.com)
Gangadhar Taduri (gangadhar3@yahoo.co.in)

Version: 1 Date: 09 Jan 2020

Author’s response to reviews:

CONSORT checklist added